Author's response to reviews

Title: Successful transureteropyeloneostomy after heminephrectomy of a double-hydronephrotic horse shoe kidney

Authors:

Holger Gerullis (holger.gerullis@gmx.net)
Christoph Eimer (christoph.eimer@web.de)
Dietmar Betz (dietmarbetz@hotmail.com)
Thomas Otto (thomas_otto@lukasneuss.de)

Version: 2 Date: 20 November 2007

Author's response to reviews:

manuscript has been translated to JMCR, submission had to be adapted